Author's response to reviews

Title: The co-administration of quetiapine or placebo to cognitive-behavior therapy in treatment refractory depression; a preliminary trial

Authors:

Yves Chaput (yveschaput@bellnet.ca)
Annick Magnan (annickmagnan@videotron.ca)
Alain Gendron (alaingendron@astrazeneca.ca)

Version: 6 Date: 12 August 2008

Author's response to reviews: see over
To the Editorial Staff,

BMC – Psychiatry,

MS: 8312126711678731

Montreal, August 11th, 2008.

Thank you for reviewing and accepting my manuscript entitled ‘The co-administration of quetiapine or placebo to cognitive-behavior therapy in treatment refractory depression; a preliminary trial’ that I re-submitted for publication in BMC-Psychiatry last week.

The corrections mentioned in your email of August the 6th have been made. I have removed the JA to my name in the web submission. I have also changed one word in the Abstract section, (with lithium or an atypical antipsychotic) rather than (lithium or with an atypical antipsychotic) at line 3. Some minor modifications to the Methods section were made that in no way affect content. A few rewordings were made in the Results section. Once again, without content modification.

In addition, several references were corrected.

In the hope that this revision will meet with your satisfaction,

________________________
Yves Chaput, MD, FRCP(C), PhD